Beam halo and beam-gas background

PYTHIA+GEANT simulation, very preliminary

Integrated rates in MF4

outer forward muon station, 150 m^2

beam-beam at L=10^{34}: 10^8 Hz

beam-gas beam-b.pipe 10^3 Hz/m 10^7 Hz along 200m at 35 and 49m

straight section max. of β oscyl.

c,b -> μ 2.10^3 10^4 π,K -> μ 10^3 5.10^3 punchthrough - 5.10^3 second. inter. - see below

Conclusions:

- rates from primary interaction are 4 orders of magnitude below the beam-beam rate - secondary interactions of particles produced in quads are more danger